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FarDet: Data-Taking
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Partition 1 (status last Week)

= diblocks 3 - 14

= Runninb

" ALL COOLED, HYV at full gain.

Kjaroqla\' Zalesak - AEM

g ~99.5% active,

Aug 25,2014

Partition 1

DB14 DB13 DB12 DB11 DB10 DB09 DB08 DB07 DB06 DB0S DB04 DB03 DB02 DBO1
TR

Current status - Retrofitting:

" APD removed from DiB-01-side dcms 07-11
= Continues with first dcm on DiB-02-side.

= Plan for this week to remove all on sides of
these two diblocks and continue with APDs
on top of DiB-03.

"P2: Diblock 01,02-top
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/FD: DAQ Uptime B

Fa SR VLN
1 @ e < [
< o
@® ] o P <
0.9 - i ® . (=] =) ®
| -
08 — Yo
< - @ e
0.7 +
@
_E 0.6 ®
g @ ®
= @ Daily
0.5 @
2 — Weekly Average
% - —4 Week Average
=04 =]

(=
W
©

0.2
0.1
0 - - -
28.7.2014 4.8.2014 11.8.2014 18.8.2014
> Last week we worse one, 82% of uptime (4 weeks average 80%).
> Beginning of the last week was very good even we started the retrofitting at that time.
> OnThu during planned beam shutdown we did many planned replacements on un-
cooled APDs or failed electronics.
» Fri beam down used to swapped of LV power supply, had also dry gas alarm issue.

» Sat & Sun we faced to other two water leaks and many issues related to them.
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FD: POTs delivered/recorded
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» The beam shutdown on Thu and Fri was down also, we used it with 86% efficiency.
» We recorded in average 0.90e18 (from 1.05) pots/ day with.
> We efﬁciently used (un—) planned beam shutdowns onThu & Fri.
> Efficiency went down during weekend when 2 water leaks occurred.
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NearDet: Detector Assembly Progress
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v Running with full HV setting (at gain of 100) and
everything cooled.

v Switched to partition 1 as all was commissioned. e
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Summary £

AT A
FarDet:

e We started a new phase: the retrofitting.

e Removed all APDs (384) on side of DiB-01 and started
removing on side of diblock-02.

NearDet:

e Near detector Completed and running within final
condition, switched to ‘Physics’ partition.

° Swapped several failed APDs.

Operation:

© Trying to get as much as possible physics data from both
detectors till next big shutdown starts.
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